Author's response to reviews

Title: Oral health status of adults in Southern Vietnam - a cross-sectional epidemiological study

Authors:

Thoa C Nguyen (nc.thoa@yahoo.com)
Dick J Witter (d.witter@dent.umcn.nl)
Ewald M Bronkhorst (e.bronkhorst@dent.umcn.nl)
Nhan B Truong (tbnhan56@yahoo.com)
Nico HJ Creugers (n.creugers@dent.umcn.nl)

Version: 3 Date: 2 March 2010

Author's response to reviews:

The following sentence has been included in the M&M section:

The Educational Scientific Committee of Cantho University of Medicine & Pharmacy granted ethics approval for this study.